Author’s response to reviews

Title: Absorbable suture can be effectively and safely used to close the mesenteric defect in a Gastric bypass Sprague-Dawley Rat Model

Authors:
Libin Yao (yaolibin_123@126.com)
Ponnie Dolo (ponnie85@yahoo.com)
Yong Shao (shaoyutong1975@163.com)
Chao Li (18068718716@189.cn)
Jason Widjaja (drjwbs@yahoo.com)
Jian Hong (hongjiajian@qq.com)
Xiaocheng Zhu (zhuxccf@163.com)

Version: 1 Date: 29 Sep 2019

Author’s response to reviews:

In this study, we demonstrated that absorbable suture can be effectively and safely used to close mesenteric defect through our animal experiment. We have made the changes to the manuscript based on the comments and recommendations from the reviewer.